ORIGINAL ARTICLE



Long-term outcomes in patients with COPD treated with non-invasive ventilation for acute hypercapnic respiratory failure

Shane O'Brien¹ · Cara Gill¹ · Niall Cograve² · Mark Quinn¹ · Ruairi J. Fahy¹

Received: 2 February 2024 / Accepted: 12 April 2024 $\ensuremath{\mathbb{O}}$ The Author(s) 2024

Abstract

Purpose Chronic Obstructive Lung Disease (COPD) remains a major cause of morbidity and mortality across the world. We evaluated survival over 9 years in a cohort of patients with COPD requiring acute inpatient non-invasive ventilation (NIV). We analyzed prognostic indices to evaluate if they were associated with mortality.

Patients and methods We performed a retrospective chart review of all patients who were admitted to St. James's Hospital respiratory ward with COPD and acute hypercapnic respiratory failure who required NIV over a 12-month period and followed their outcomes over 9 years. We investigated the association between survival and potential prognostic variables using univariate analysis and multivariate Cox proportional hazards model. We evaluated the association between survival and the following parameters: age, gender, multiple admissions requiring NIV (> 1 admission in within 12 months of index presentation), home NIV use preadmission, initial arterial blood gas pH, days spent on NIV, serum albumin and serum albumin to serum CRP ratio at admission.

Results Ninety-nine patients with COPD and acute hypercapnic respiratory failure were identified over a 12-month period from January to December 2011. Survival at 1, 2, 5 and 9 years was 65% (n=64), 42% (n=42), 25% (n=25) and 21% (n=21), respectively. Increasing age (p value < 0.001) and a lower serum albumin (p value < 0.005) were associated with a higher mortality. There was a trend towards improved survival in the group who were treated with home NIV prior to admission compared to no NIV therapy at home but this did not reach statistical significance (Fig. 3, p value = 0.088).

Conclusion This study highlights the long-term mortality in patients with COPD admitted with hypercapnic respiratory failure requiring NIV and correlates with prior studies. Increasing age and lower serum albumin were associated with increased mortality. Home NIV may have a protective long-term survival benefit in patients with COPD who have been admitted for acute NIV.

Keywords Chronic Obstructive Pulmonary Disease · Home non-invasive ventilation · Mortality · Predictors of survival

Plain Language Summary

This study highlights the poor long-term survival in patients with COPD admitted with hypercapnic respiratory failure requiring non-invasive ventilation (NIV, a form of respiratory support via mask). Increasing age and lower serum albumin were associated with increased mortality. Home NIV (NIV used by patients at home) may have a protective long-term survival benefit in patients with COPD who have been admitted for acute NIV.

Shane O'Brien Shaneobrien2@hotmail.com

Introduction

Despite declining rates of smoking in Europe and North America, Chronic Obstructive Lung Disease (COPD) remains a major cause of long-term morbidity and mortality with frequent hospitalizations adding to the cost of care [1, 2]. Patients admitted with acute exacerbations have a substantial hospital mortality rate [1]. Currently, the treatment of choice for hypercapnic respiratory failure is non-invasive mechanical ventilation (NIV) given its proven reduction in mortality [3]. Many of these studies were done in the ICU setting, though it is now common practice to manage these patients on a dedicated respiratory ward and little data is available on long-term survival from ward-based NIV services outside of a trial setting [1, 2].

¹ Department of Respiratory Medicine, St. James' Hospital, Dublin, Ireland

² School of Medicine, Trinity College Dublin, Dublin, Ireland

Additionally, clinical or biological variables that could predict long-term survival may prove useful in a real-world setting. This is especially the case relating to home NIV, as recent evidence suggests that for those patients with persistent hypercapnia following an acute exacerbation of COPD, adding home non-invasive ventilation to home oxygen therapy prolonged the time to readmission or death within 12 months [4].

In this regard, we evaluated survival over a 9-year period in a cohort of patients, the majority of whom have COPD after an acute inpatient admission requiring NIV. We also analyzed prognostic indicators to evaluate if they were associated with mortality.

Material and methods

St James's Hospital is a large tertiary referral hospital in Dublin with a dedicated respiratory NIV unit since 2003. We performed a retrospective chart review of all patients who were admitted to St. James's Hospital Respiratory ward with acute hypercapnic respiratory failure who required NIV over 12 months and followed their outcomes over 9 years. The primary endpoint was overall survival. Patients were treated with NIV if they met the selection criteria. At the time of study in our facility, this included an appropriate diagnosis of hypercapnic respiratory failure with potential reversibility requiring two of the following parameters:

- 1. Moderate to severe respiratory distress, tachypnoea, accessory muscle use
- 2. Arterial blood gas showing pH < 7.35, $paCO_2 > 6$ kPa

Data retrospectively collected included demographics, comorbid medical illnesses, laboratory investigations, radiology and arterial blood gases in the acute setting, both initially and during treatment. We investigated the association between survival and potential prognostic variables using univariate analysis. The multivariate Cox proportional-hazards model was used to control for confounding factors. We adjusted the association between home NIV use, serum albumin and serum albumin to CRP ratio for age using the multivariate Cox proportional-hazards model. We evaluated the association between overall survival and the following parameters: age, gender, multiple admissions requiring NIV (>1 admission within 12 months of index presentation), home NIV use pre-admission, initial arterial blood gas pH, days spent on NIV, serum albumin and serum albumin to serum CRP ratio at admission. Missing data was predominantly handled in the study design and with pairwise deletion of missing at random data. We had intended to assess the association between pulmonary function test parameters; however, due to an unacceptable level of missing PFT results, we decided to omit this from our analysis. Poisson regression was used to evaluate if any of the variables in our study were associated with the number of acute admissions requiring NIV. We attempted to minimize selection bias by including all patients who fulfilled the above selection criteria in the predefined time period. Patients with a diagnosis of obstructive sleep apnea or obesity hypoventilation syndrome were excluded as we feel this is a separate pathology which also causes hypercapnic respiratory failure. There is a ward-based system for capturing the diagnoses of each patient who is starting NIV. COPD was diagnosed by a respiratory physician in St. James's Hospital and patients with a diagnosis of COPD treated with acute NIV were recorded in a ward-based database. We excluded four patients who had a diagnosis of obstructive sleep apnea. Potential sources of selection bias included patients who may have been too critically unwell to be deemed appropriate for a trial of NIV and therefore were not captured in this study. For instance, patients who have a ward-level ceiling of care, who in other circumstances require intubation, may have been treated with comfort measures rather than a trial of NIV depending on the specific circumstances.

This study involves human participants and was approved by the St James Hospital/Tallaght University Hospital Joint Research Ethics Committee, R&I number 706, submission number 168. This study complies with the declaration of Helsinki.

Results

Ninety-nine patients were identified over 12 months from January to December 2011. Patient characteristics and results are summarized in Table 1.

There was an equal gender balance with female patients representing 55% (n = 54) of the cohort and 45% (n = 45) were male. The median age at admission was 67 years (range 29–93). There was no difference between males and females concerning survival, home NIV use, or death on index admission.

The median length of stay was 9 days (range 1–34). Twenty-seven percent of patients (n=27) were on home NIV. Ninety-six percent of patients (n=96) had a confirmed diagnosis of COPD.

Overall survival at 1, 2, 5 and 9 years was 65% (n = 64), 42% (n = 42), 25% (n = 25) and 21% (n = 21), respectively.

Table 1 Patient characteristics and results

n	99
Age (mean (SD))	67.7 (15.9)
Sex, <i>M</i> (%)	45 (45.5)
Number of patients with COPD diagnosis	96 (96%)
Home NIV, <i>n</i> (%)	27 (27.3%)
1-year survival	65% (n = 64),
2-year survival	42% (n=42)
5-year survival	25% (n=25)
9-year survival	21%(n=21)
Number of patients with > 1 admission for acute NIV within 12 months from index presenta- tion (%)	21 (21.2%)
Time to death in days (mean (SD))	621 [74.0, 2163.5]
Days on NIV (median [IQR])	8 [5., 13.]
Death during index admission (%)	24 (24.2%)
Admission serum albumin g/L (mean (SD))	34.55 (5.33)
Admission ABG pH (mean (SD))	7.31 (0.08)
Admission pCO2 (mean)	8.00

A total of 21.9% (n=21) had recurrent admissions (>1 within 12 months of index presentation requiring NIV (range between 2 and 5 admissions). Patients with multiple admissions for non-invasive ventilation (>1 admission within 12 months of index presentation) had a lower 9-year survival than those requiring only one admission; however, there was no significant association between recurrent admissions and survival (p value =0.7) (Fig. 1).

Survival at 1, 2 and 5 years for patients requiring > 1 admissions for NIV in the first year of study was 50% (n=8), 37.5% (n=6) and 25% (n=4), respectively.

The small cohort of patients requiring > 2 admissions in the first year of the study (range 3–6 admissions) had a survival at 1, 2 and 5 years of 60% (n=3), 20% (n=1) and 0% (n=0), respectively.

Poisson regression was utilized to evaluate if any of the variables in our study were associated with the number of admissions requiring NIV. There was an association with

 Table 3
 Univariate analysis and Cox proportional hazard models

Variable	HR	95% Confidence intervals	р
Age	1.05	1.03-1.06	< 0.01
Sex	1.11	0.71-1.72	0.65
Days on NIV	0.99	0.96-1.01	0.63
Greater than one admission for acute NIV (%)	1.11	0.65-1.88	0.69
Home NIV	0.64	0.38-1.07	0.09
Admission serum albumin	0.96	0.92-0.99	0.03
Admission pH	0.12	0.003-3.44	0.21
CRP	1.01	0.99-1.01	0.17
Albumin to CRP ratio	0.97	0.95-0.99	0.03

home NIV being protective with regard to the number of admissions (Table 4).

Age was the greatest predictor of survival at index presentation, with a median age of 80.2 years amongst those who died compared to 63.6 years amongst those who survived to discharge (p < 0.001) (Table 2 and Fig. 2).

Serum albumin was considerably lower in the group that died at index presentation (mean serum albumin 32.0 g/L, p < 0.005) compared to those that survived to discharge (mean serum albumin 35.4 g/L, p < 0.005) (Table 2).

Using the Cox proportional hazard model, we showed that increasing age (age hazard ratio [HR], 1.05; confidence intervals [CI], 1.03–1.067; *p* value < 0.001) and lower serum albumin (albumin: HR, 0.96; CI 0.92–0.97; *p* value = 0.03) were associated with lower 9-year survival (Table 3).

There was no statistically significant association between sex, multiple admissions requiring NIV (> 1 admissions within 12 months of index presentation), initial arterial blood gas pH, days on NIV, or preexisting home NIV use and 9-year survival. When adjusted for age using multivariate analysis, albumin, albumin to CRP ratio and home NIV use did not produce a significant p value with respect to predicting survival.

Table 2 Results stratified bysurvival to discharge versusdeath on index admission

	Survived to discharge at index admission	Died at index admission	р
n	75	24	
Age (mean (SD))	63.6 (15.3)	80.2 (10.2)	< 0.01
Sex, <i>M</i> (%)	35 (46.7)	10 (41.7)	0.85
Time on NIV in days (median [IQR])	8.0 [5.0, 12.7]	8.50 [4.7, 14.0]	0.94
Home NIV, $N(\%)$	23 (30.7)	4 (16.7)	0.28
Alb, g/L (mean (SD))	35.4 (5.0)	31.9 (5.5)	< 0.01
ph (mean (SD))	7.3 (0.08)	7.28 (0.07)	0.08



Fig. 1 Kaplan-Meier survivor function stratified by multiple admissions (>1 admission within 12 months of presentation) requiring non-invasive ventilation. Red line represents patients with just one admission in the first year and blue line represents patients with greater than one admission in the first year. Survival probability % on *y* axis and time in days on *x* axis

There was a trend towards improved 9-year survival in the group who had pre-existing home NIV before admission compared to no NIV therapy at home (Fig. 3, pvalue = 0.088). Although there is a trend toward a positive finding here, patients on home NIV were younger (mean age, 61.85 years versus 69.88 years) (Table 3).

On reviewing data 9 years on from the index admission, 21 patients (21%) were still alive and 8 of these are on home NIV, suggesting a trend toward NIV having a protective effect (p=0.11) (Table 4).

Discussion

The early use of NIV for patients with COPD and hypercapnic respiratory failure in the general ward setting leads to a improvement of physiological variables, a reduction in the need for invasive mechanical ventilation and a reduction in in hospital mortality [5].

The 1-, 2-, 5- and 9-year survival figures in our study of patients treated with NIV were 65% (n=64), 42% (n=42), 25% (n=25) and 21% (n=21), respectively, which were similar to what has been reported in the literature to date.

Cabrini et al. reported a 1-year survival of 66% in patients with COPD receiving ward-based non-invasive ventilation for acute respiratory failure [6]. Titlestad et al. reported a 5-year survival of 23% in patients receiving NIV for the first time after being admitted acutely to an acute medical ward with later transfer to a respiratory ward with respiratory failure and a diagnosis of COPD [7].

Chung et al. found that the long-term survival of patients surviving the first episode of type II respiratory failure requiring non-invasive ventilation (NIV) at 1, 2 and 5 years was 72%, 52% and 26%, respectively [8]. In their study, only advanced age (p=0.04), BMI (p=0.014) and prior domiciliary oxygen use (p=0.03) correlated with death within 5 years. Physiological measures of the severity of respiratory failure at presentation did not predict survival and this is in keeping with our findings in this study. In patients admitted to hospital with an exacerbation of COPD, Sprooten and colleagues identified advanced age, NIV use more than 8 days and unsuccessful response to NIV as clinical important independent predictors for long-term mortality [9]. Age was the greatest predictor of survival in our study, and we did not find an association between the severity of acidosis or time spent on NIV at presentation and survival. In our study, lower serum albumin was associated with a worse survival. It is known that serum albumin reflects nutritional status which may account for the association. However, other stressors can reduce serum albumin [10]. Serum albumin is depressed during inflammation, and thus, it is intuitive that patients presenting with exacerbations of COPD or infection may have lower serum albumin [11]. In our study, lower serum albumin was associated with worse survival overall and this may be related to the patient's nutritional status and/ or levels of inflammation [12].

Table 4Results stratified bywhether patient was on homeNIV before admission or not

	Not prescribed home NIV	Prescribed home NIV	р
n	72	27	
Age (mean (SD))	69.88 (15.5)	61.85 (15.6)	0.03
Gender, $M(\%)$	33 (45.8)	12 (44.4)	1
Time to death in days (median [IQR])	579 [33.5, 1419.]	868 [320, 3000.]	0.07
Time on NIV in days (median [IQR])	7.00 [4.5, 13.0]	8.0 [5.0, 12.5]	0.42
Death on index presentation (%)	20 (27.8)	4 (14.8)	0.28
Greater than one admission for acute NIV	13 (18.1)	8 (29.6)	0.33
Alb, g/L (mean (SD))	34.4 (4.9)	34.9 (6.3)	0.64
ph (mean (SD))	7.3 (0.08)	7.3 (0.08)	0.40



Fig. 2 Kaplan-Meier survivor function stratified by age tertiles. Survival probability % on *y* axis and time in days on *x* axis. Red line represents patients less than 64 years old. The green line represents patients aged 64–75 years old. The blue line represents patients above 75 years old

There was no significant association with recurrent admissions in the first year and subsequent survival over 9 years in our study. This may be due to survivor bias; one needs to be alive to have a subsequent admission, and therefore, those who died early in the study did not contribute to the mortality associated with recurrent admissions. Also, the number of patients with multiple admissions was relatively small (n = 21) and this too may have impacted the statistical power.

Regarding the use of home NIV in patients with COPD and hypercapnia, evidence is accumulating from both



Fig. 3 Kaplan-Meier survivor function stratified by the use of non-invasive ventilation (NIV) at home before admission. Red line represents patients not on home NIV and blue line represents patients on home NIV. Survival probability % on *y* axis and time in days on *x* axis

metanalysis and randomized controlled clinical trials that home NIV may reduce admissions and improve survival. Wilson and colleagues performed a meta-analysis of patients with COPD and hypercapnia and found that home NIV, compared with no device, was associated with lower risk of mortality (22.31% vs 28.57%; risk difference [RD], -5.53% [95% CI, -10.29 to -0.76%]; odds ratio [OR], 0.66 [95% CI, 0.51–0.87]; *p* = .003; 13 studies; 1423 patients; strength of evidence [SOE], moderate), hospital admissions and intubation [13]. The American Thoracic Society guidelines now suggest the use of nocturnal NIV in addition to usual care for patients with chronic stable hypercapnic COPD [14]. Our data show a trend towards improved survival over 9-year follow-up in patients using home NIV, and this adds to the growing body of evidence for the use of home NIV in patients with COPD and hypercapnia. This study was a retrospective, single-centre observational study without a comparator group and results need to be taken within this context; however, despite this, we feel that it adds value to the field. Survival in COPD is poor overall. Five-year survival in patients with COPD was 79% compared to 95% survival in controls with unobstructed pulmonary function tests matched for age, comorbidities and smoking history [15]. In a recent study which aimed to determine whether long-term NIV (LT-NIV) initiated after an admission with acute hypercapnic respiratory failure (AHRF) can affect survival and admission rate in COPD patients, time to readmission with AHRF or death within 12 months was numerically smaller in the LT-NIV group, however did not reach significance [16]. The poor 9-year survival in our cohort once likely reflects the high mortality associated with exacerbations causing hypercapnic respiratory failure in patients with COPD. The cohort of patients with more exacerbations leading to hypercapnic respiratory failure had a lower survival; however, the numbers in this study were likely too small to produce a significant result.

Conclusion

This study highlights the long-term mortality in patients with COPD admitted with hypercapnic respiratory failure requiring NIV and correlates with prior studies. Increasing age and lower serum albumin are associated with increased mortality. Home NIV may have a protective long-term survival benefit in patients with COPD who have been admitted for acute NIV. Serum albumin could also be a potentially useful predictive biomarker of survival in this context. Author contributions statement All authors made a significant contribution to the work reported, whether that is in the conception, study design, execution, acquisition of data, analysis and interpretation, or in all these areas; took part in drafting, revising or critically reviewing the article; gave final approval of the version to be published; have agreed on the journal to which the article has been submitted; and agree to be accountable for all aspects of the work.

Funding Open Access funding provided by the IReL Consortium

Data availability Data is available from the corresponding author upon request.

Declarations

Ethics approval and consent to participate This study involves human participants and was approved by the St James Hospital/ Tallaght University Hospital Joint Research Ethics Committee, R&I number 706, submission number 168. As this was a pseudonymized retrospective chart review, individual consent was not required nor obtained. No identifiable details relating to patients were included in the study. This study complies with the declaration of Helsinki.

Competing interests The authors declare no competing interests.

Patient and public involvement statement The main outcome of this study is survival and predictors of survival, and we feel this is patient-centred. Patients were not directly involved in the design of this study.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

References

- 1. Müllerova H, Maselli DJ, Locantore N et al (2015) Hospitalized exacerbations of COPD: risk factors and outcomes in the ECLIPSE cohort. Chest 147(4):999–1007. https://doi.org/10. 1378/chest.14-0655
- Ehteshami-Afshar S, FitzGerald JM, Doyle-Waters MM et al (2016) The global economic burden of asthma and chronic obstructive pulmonary disease. Int J Tuberc Lung Dis 20:11–23. https://doi.org/10.5588/ijtld.15.0472
- 3. Davidson AC, Banham S, Elliott M et al (2016) BTS/ICS guideline for the ventilatory management of acute hypercapnic respiratory failure in adults. Thorax 71(Suppl 2):ii1-35
- Murphy PB, Rehal S, Arbane G et al (2017) Effect of home noninvasive ventilation with oxygen therapy vs oxygen therapy alone on hospital readmission or death after an acute COPD

exacerbation: a randomized clinical trial. JAMA 317(21):2177–2186. https://doi.org/10.1001/jama.2017.4451

- Plant PK, Owen JL, Elliott MW (2000) Early use of non-invasive ventilation for acute exacerbations of chronic obstructive pulmonary disease on general respiratory wards: a multicentre randomised controlled trial. Lancet 355(9219):1931–1935. https:// doi.org/10.1016/s0140-6736(00)02323-0
- Cabrini L, Landoni G, Bocchino S et al (2016) Long-term survival rate in patients with acute respiratory failure treated with noninvasive ventilation in ordinary wards. Crit Care Med 44(12):2139–2144. https://doi. org/10.1097/CCM.00000000001866
- Titlestad IL, Lassen AT, Vestbo J (2013) Long-term survival for COPD patients receiving noninvasive ventilation for acute respiratory failure. Int J Chron Obstruct Pulmon Dis 8:215–219. https:// doi.org/10.2147/COPD.S42632
- Chung LP, Winship P, Phung S et al (2010) Five-year outcome in COPD patients after their first episode of acute exacerbation treated with non-invasive ventilation. Respirol 15(7):1084–1091. https://doi.org/10.1111/j.1440-1843.2010.01795.x
- Sprooten RTM, Rohde GGU, Janssen MTHF et al (2020) Predictors for long-term mortality in COPD patients requiring non-invasive positive pressure ventilation for the treatment of acute respiratory failure. Clin Respir J 14(12):1144–1152. https:// doi.org/10.1111/crj.13251. Epub 2020 Sep 22. PMID: 32780940; PMCID: PMC7756413
- Rothschild MA, Oratz M, Schreiber SS (1988) Serum albumin. Hepatology 8(2):385–401. https://doi.org/10.1002/hep.1840080234
- 11. Banh L (2006) Serum proteins as markers of nutrition: what are we treating? Pract Gastroenterol 30(10):46
- Moshage HJ, Janssen JA, Franssen JH et al (1987) Study of the molecular mechanism of decreased liver synthesis of albumin in inflammation. J Clin Invest 79(6):1635–1641. https://doi.org/10. 1172/JCI113000
- Wilson ME, Dobler CC, Morrow AS et al (2020) Association of home noninvasive positive pressure ventilation with clinical outcomes in chronic obstructive pulmonary disease: a systematic review and meta-analysis. JAMA 323(5):455–465. https://doi.org/10.1001/jama.2019.22343.PMID:32016309; PMCID:PMC7042860
- Macrea M, Oczkowski S, Rochwerg B et al (2020) Long-term noninvasive ventilation in chronic stable hypercapnic chronic obstructive pulmonary disease. An official American Thoracic Society clinical practice guideline. Am J Respir Crit Care Med 202(4):74–87. https://doi.org/10.1164/rccm.202006-2382ST. PMID: 32795139; PMCID: PMC7427384
- Miniati M, Monti S, Pavlickova I, Bottai M (2014) Survival in COPD: impact of lung dysfunction and comorbidities. Medicine (Baltimore) 93(12):e76. https://doi.org/10.1097/MD. 0000000000000076
- Hedsund C, Linde Ankjærgaard K, Peick Sonne T et al (2023) Long-term non-invasive ventilation for COPD patients following an exacerbation with acute hypercapnic respiratory failure: a randomized controlled trial. Eur Clin Respir J 10(1):2257993. https://doi.org/10.1080/20018525.2023.2257993.PMID:37746 028;PMCID:PMC10512815

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.